[tex54]What is wrong with these coexistence lines?

The figure shows the two-phase coexistence lines near the triple point of a hypothetical substance. Under the reasonable assumption that the densities of the liquid and solid phases are similar and much larger than the density of the gas phase, this phase diagram implies a material property that is very pathological under the circumstance described.

Identify the aspect of the phase diagram that is suspect. Identify the pathological property of the substance that follows from this phase diagram.

Hint: a somewhat related exotic property is observed in superfluid^3He.